Author’s response to reviews

Title: Dental hygienists’ perceptions of professionalism are multidimensional and context-dependent: A qualitative study in Japan

Authors:

Yukiko Nagatani (nagayuki@hiroshima-u.ac.jp)
Rintaro Imafuku (rimafuku@gifu-u.ac.jp)
Toshinobu Takemoto (takefn@hiroshima-u.ac.jp)
Tadayuki Waki (wtadayuki@gmail.com)
Taiji Obayashi (Taiji@hiroshima-u.ac.jp)
Tetsuji Ogawa (tbach@hiroshima-u.ac.jp)

Version: 2 Date: 13 Dec 2017

Author’s response to reviews:

Dear Editor,

On behalf of my co-authors, I would like to thank you for your comments on our manuscript (MEED-D-17-00304) entitled ‘Dental hygienists’ perceptions of professionalism are multidimensional and context-dependent: A qualitative study in Japan’. We would also like to thank the reviewers for their really positive comments on our paper.

We have asked a professional language service for editing and correcting the grammar, wording, spelling, and punctuation in this manuscript. We hope that the corrections would be satisfactory, and you would accept our paper for publication in BMC Medical Education.

Sincerely,

Rintaro Imafuku
Gifu University
Medical Education Development Center